

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sharma, Rasadhika; Nguyen, Thanh Tung; Grote, Ulrike; Nguyen, Trung Thanh

Working Paper Changing livelihoods in rural Cambodia: Evidence from panel household data in Stung Treng

ZEF Working Paper Series, No. 149

Provided in Cooperation with: Zentrum für Entwicklungsforschung / Center for Development Research (ZEF), University of Bonn

Suggested Citation: Sharma, Rasadhika; Nguyen, Thanh Tung; Grote, Ulrike; Nguyen, Trung Thanh (2016) : Changing livelihoods in rural Cambodia: Evidence from panel household data in Stung Treng, ZEF Working Paper Series, No. 149, University of Bonn, Center for Development Research (ZEF), Bonn

This Version is available at: https://hdl.handle.net/10419/144856

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Working Paper 149

Rasadhika Sharma, Thanh Tung Nguyen, Ulrike Grote, Trung Thanh Nguyen

Changing Livelihoods in Rural Cambodia: Evidence from panel household data in Stung Treng



ZEF Working Paper Series, ISSN 1864-6638 Center for Development Research, University of Bonn Editors: Editors: Christian Borgemeister, Joachim von Braun, Manfred Denich, Solvay Gerke, Eva Youkhana and Till Stellmacher

Authors' addresses

Rasadhika Sharma Institute for Environmental Economics and World Trade Leibniz University Hannover (LUH) Königsworther Platz 1 30171 Hannover, Germany Tel. 0049 (0)511-762-4087; Fax 0511-762-2667 E-mail: sharma@iuw.uni-hannover.de www.iuw.uni-hannover.de

Thanh Tung Nguyen Institute for Environmental Economics and World Trade Leibniz University Hannover (LUH) Königsworther Platz 1 30171 Hannover, Germany Tel. 0049 (0)511-762-14575; Fax 0511-762-2667 E-mail: tung.nguyen@iuw.uni-hannover.de www.iuw.uni-hannover.de

Prof. Dr. Ulrike Grote ZEF Senior Fellow Institute for Environmental Economics and World Trade Leibniz University Hannover (LUH) Königsworther Platz 1 30171 Hannover, Germany Tel. 0049 (0)511-762-4185; Fax 0511-762-2667 E-mail: grote@iuw.uni-hannover.de www.iuw.uni-hannover.de

PD Dr. Trung Thanh Nguyen Institute for Environmental Economics and World Trade Leibniz University Hannover (LUH) Königsworther Platz 1 30171 Hannover, Germany Tel. 0049 (0)511-762-4827; Fax 0511-762-2667 E-mail: thanh.nguyen@iuw.uni-hannover.de www.iuw.uni-hannover.de

Changing Livelihoods in Rural Cambodia

Evidence from panel household data in Stung Treng

Rasadhika Sharma, Thanh Tung Nguyen, Ulrike Grote, Trung Thanh Nguyen

Abstract

Analysis of livelihood strategies can aid to understand and resolve problems associated with vulnerability to poverty and food security. This paper aims to identify and describe the changes in rural livelihood activities by using household data for 2013 and 2014 collected in Stung Treng, Cambodia. We use the same variables and estimate different clusters for both the years. The paper concludes that despite the lag of only one year, there are noticeable changes in livelihood strategies. Firstly, we find a group of transition farmers in 2014 that is composed of households that are witnessing a shift towards commercialization. They invest and consume more than subsistence farmers. Secondly, there is a greater diversification in activities amongst the groups. Most households practice multiple activities. Lastly, with regards to self-employment, there has been a shift from agriculture and the production sector to services and crafts. All of the above changes can be deemed as positive as there is a gradual movement away from more vulnerable sectors. Accordingly, households that participate in livelihood activities related to agriculture and natural resource extraction are most affected by shocks and face the highest vulnerability to poverty. The paper additionally highlights some concerns such as a decline in availability of extracted products such as the fish stock which are expected to negatively impact on these more vulnerable rural households in the medium and longer term. Furthermore, the state of education is dismal and needs attention. Therefore, policy makers need to consider these issues while addressing rural poverty.

Keywords: Livelihoods, Rural poverty, Diversification, Cluster Analysis, Cambodia

JEL classification: Q57, Q20, Q12

Content

AB	ABSTRACT				
СС	NTENT	r	ш		
LIS	T OF T	ABLES	IV		
1	INTR	DDUCTION	1		
	1.1	Problem statement	1		
	1.2	Research objectives	2		
2	DATA	N N N N N N N N N N N N N N N N N N N	3		
	2.1	Study site and data collection	3		
	2.2	Questionnaire design	3		
3	LIVEL	IHOOD CLUSTERS AND THEIR CHARACTERISTICS IN STUNG TRENG	4		
	3.1	Identifying relevant livelihood clusters	4		
	3.2	Changes in livelihood strategies	5		
	3.3	Characteristics of the clusters	6		
	3.3.1	Income and consumption	6		
	3.3.2	Poverty	/		
4	SELEC	CTED LIVELIHOOD ACTIVITIES AND THEIR DETERMINANTS	9		
	4.1	Agriculture	9		
	4.1.1 4 1 2	Farming Livestock	9 10		
	4.2	Fishing hunting and logging	10		
	4.2.1	Extracted products	11		
	4.2.2	Income contribution	12		
	4.3	Business and wage employment	13		
	4.3.1	Self-employment	13		
	ч. <u>э</u> .2 Л Л	Migration	14		
	4.4.1	Destination of migrants from Stung Treng	15		
	4.4.2	Types of jobs in destination areas	15		
	4.4.3	Migrant remittances and household welfare	16		
5	DEM	OGRAPHIC DIVIDEND AND SHOCKS	17		
	5.1	Demographic dividend	17		
	5.2	Shocks	18		
6	CONC	CLUSION	20		
7	BIBLI	OGRAPHY	22		
8	APPE	8 APPENDIX			

List of Tables

Table 3-1 Classification of livelihoods by cluster	4
Table 3-2 Changes in Clusters and their Livelihood Strategies (No. of households and in %)	5
Table 3-3 Income and consumption by cluster (in USD PPP)	6
Table 3-4 Poverty (headcount ratio) based on income/consumption by cluster	7
Table 4-1 Average agricultural characteristics for rice (mean)	9
Table 4-2 Livestock rearing in Stung Treng (mean)	10
Table 4-3 Extraction of natural resources	11
Table 4-4 Contribution of natural resources to annual household income	12
Table 4-5 Self-employment by sector	13
Table 4-6 Off-farm employment by sector	14
Table 4-7 Migrant distribution by destination	15
Table 4-8 Types of occupation in destination areas	15
Table 4-9 Migrant remittances and household welfare	16
Table 5-1 Educational statistics (16-60 years)	17

Table 5-1 Educational statistics (16-60 years)	17
Table 5-2 Shocks by categories	18
Table 5-3 Shocks by cluster	19
Table 5-3 Shocks by cluster	19

Table 8-1 Variable list and summary statistics for clusters 2013	25
Table 8-2 Variables and summary statistics 2014	26
Table 8-3 Average agricultural characteristics by cluster for rice (mean)	27
Table 8-4 Average value (mean) of livestock rearing for households by cluster	28
Table 8-5 Property rights enforcement status of the extracting grounds	28
Table 8-6 Selected socio-demographic characteristics of migrants (%)	28
Table 8-7 Shocks frequency by cluster	29

1 Introduction

1.1 **Problem statement**

The World Bank lists Cambodia in the group of Least Developed Countries. Though deemed as one of the fastest growing economies in Asia with an annual growth rate of 7.8% on average between 2000 and 2014, the country still has a relatively low Gross Domestic Product per capita of \$1,095¹. The poverty rate has declined significantly from 48% in 2007 to 18% in 2012 (ADB, 2015). However, most Cambodians are still vulnerable to fall into poverty. According to the World Bank (2014), a vast majority of the population that escaped poverty was able to do it only marginally and hence, there is a huge share of near poor in the country.

Interestingly, most of these near poor and poor households are concentrated in specific livelihoods that relate to either agriculture or natural resources. Therefore, analyzing livelihood strategies could provide great insights and aid to resolve problems that are associated with vulnerability to poverty and food security. Also, it is important to understand the drivers of households adopting certain livelihood strategies and the opportunities of livelihood diversification.

Agriculture employs more than half of Cambodia's labor force while contributing only 37% to its GDP (FAO, 2014). Productivity of the sector is a big concern (Siphana et al., 2011) and there is presence of vulnerability as the production and trade portfolio is highly skewed towards rice. Production of paddy rice increased to 9.3 million tons in 2013 from 4.3 million tons in 2003. A net exporter of rice, Cambodia exported around 4 million tons alone in 2013. Also, Cambodia faces great competition from Vietnam and Thailand in rice production. Though the production cost is comparatively lower, the processing costs are much higher than for its counterparts. Additionally, while most farmers in the neighboring countries possess formal land titles, the same is not true for Cambodia (Nguyen et al., 2016).

Rural livelihoods are also greatly reliant on Cambodia's natural resources. Nguyen et al. (2015) establish in their study that environmental resources contribute significantly to the household income in Cambodia. It should be mentioned that while 60% of the land is under forest cover, there are many water resources. The Mekong River that runs from North to South is navigable for most of its course and along with the Tonle Sap provides excellent fishing and aquaculture opportunities. However, there has been a gradual decline in natural resources over the last few years. This could be attributed to many factors such as illegal and unsustainable fish and timber harvesting, that is carried out by military and local authorities and commercial enterprises (McKenny and Tola, 2002). Furthermore, the natives have a heavy dependence on these resources not only to supplement their income but also to fulfill nutritional needs.

Our case study site, Stung Treng is a remote province in Cambodia that is characterized by a huge presence of poverty and food insecurity. It is considered to be richer in natural resources than most of the other provinces (NIS, 2013). Generally, farming is small-scale and there is an absence of business opportunities other than logging and fishing. To add to this, the area has very poor infrastructure. There is no railroad and roadways do not cover a large part of the province. On the educational front, the dropout rate is 80% majorly comprising girls (Hungry for Life, 2014).

There are very few studies that have focused on rural Cambodia to analyze livelihood strategies (McKenney and Tola, 2002; FAO, 2010). By using livelihood activities to cluster households, it becomes easier to delineate households that are more prone to poverty. This study is a follow up of the paper by Bühler et al., 2015 which provided many interesting observations with relation to

¹ According to the World Bank website, Vietnam had a growth rate of 6 % between 2011 and 2015 with GDP per capita of \$2,052. While, Thailand registered a growth rate of 0.9% during 2011 and 2015 and has GDP per capita of \$5,997.

livelihood strategies. The current study endeavors to add to the existing knowledge. While the first paper was based on data from only 2013, we analyze panel data available for 2013 and 2014. Though the time period is not long yet there are certain factors that could have encouraged changes in livelihoods. Firstly, there has been a noticeable drop in the number of fish extracted by the households between the two years which indicates a decline in fish stocks. Secondly, there are possible anticipation effects associated with the construction and opening of dams such as the Stung Treng dam and the Lower Sesan Dam 1 and 2 on the perceptions of people². Thirdly, the number and severity of shocks experienced by the households vary between 2013 and 2014. Analysing this feature could add a strong dynamic aspect to the study. Lastly, there have been some minor infrastructural improvements in recent times. For example, a bridge was constructed on the Mekong in 2014³. Therefore, an in-depth analysis of the available data is pertinent and could pave way for further research.

1.2 **Research objectives**

The discussion paper aims to add to the existing knowledge on livelihood strategies with respect to households in rural Cambodia. Additionally, the paper also provides basic insights into the demographic dividend situation in the province along with shocks and coping strategies.

Specifically, it tackles the following two objectives:

- (1) To identify and describe the changes in rural livelihood strategies between 2013 and 2014,
- (2) To analyze selected livelihoods and their determinants.

The paper is organized as follows: section two describes the study site and the data collection methods. Section three describes the process of clustering. It also gives an overview of the clusters and the changes in livelihood strategies between 2013 and 2014. Section four analyses selected livelihood strategies and their determinants. Section five provides some insights into demographic dividend and shocks. Section six presents the summary and concludes.

² Stung Treng Dam will be opened in 2016 while the Lower Sesan 1 and 2 are proposed to open in 2019

³ Observation made by enumerators while surveying.

2 Data

2.1 Study site and data collection

Stung Treng is a province located in North-eastern Cambodia and is 481 km from the nation's capital, Phnom Penh. It borders Lao PDR to the North. Covering an area of 12,061 km², it comprises 5 districts with 34 communes and 129 villages. There are 95,000 inhabitants belonging to 17,900 households. The Mekong divides the state roughly into two halves and has traditionally served as the main channel of migration from North to the South (Try and Chambers, 2006).

Stung Treng has high incidence of poverty and food insecurity. Also, there is a relatively high dependence on natural resources. Hence, the province was selected as a study site to collect household data to measure the vulnerability of rural households to poverty and food insecurity.

A sampling procedure similar to Hardeweg et al. (2013), as used in the DFG FOR 756 project⁴, was designed. This is based on the guidelines provided by the UN Department of Economic and Social Affairs (UN, 2005). At the village level, data on number of households, population and rural-urban classification was available. A self-weighted sample with clustering on village level was drawn with an assumption of agro-ecological and socio-economic homogeneity

The sample of 600 households was generated in two steps. First, from the list of villages, 30 villages were sampled as Primary Sampling Units (PSUs) proportional to size (PPS). Data from Census 2008 (NIS, 2008) was used to define a listing frame in the first step. For the second step, a random draw was made from the village level listing frame to select 20 households of each PSU. The number was kept at 20 and not more due to the low number of villages in the province. Under this method, each household has a probability of 3% of getting selected. Additional replacement households were also sampled to ensure availability of extra data in case a surveyed household was deemed ineligible. However, this was observed for less than 5% of the total households that were surveyed originally.

2.2 Questionnaire design

A questionnaire was used for the survey to measure vulnerability to poverty following Hardeweg et al. (2013). To obtain information at the household level, household heads were interviewed. This questionnaire contains 89 pages and 616 variables. It includes not only basic data on household individuals but also covers various sections such as agriculture, investment, food consumption, off farm employment, shocks and borrowing and lending behaviors of the household.

⁴ For more information: https://www.vulnerability-asia.uni-hannover.de/6681.html

3 Livelihood clusters and their characteristics in Stung Treng

3.1 Identifying relevant livelihood clusters

In order to describe the changes in livelihood strategies, a separate cluster analysis was performed for both the years. A decision was made not to merge the data sets for 2013 and 2014 to control for any loss of information. The overall analysis for each year was undertaken in two steps. First, a Principal Component Analysis (PCA) was performed to reduce the number of input variables into major factors and then Cluster Analysis was used to group households according to their livelihood inputs.

Our approach makes use of input variables to map input factors for livelihood strategies. This is in contrast to studies such as Babulo et al., 2008 and de Sherbinin et al., 2008 that use income shares to identify livelihood strategies. As stochastic nature of income could induce variations in dependency upon income for different years (Nielsen et al., 2013), the input allocation approach is given a preference.

Based on the collected data, the same 22 variables are used for both the years. These refer to investment, cost, land and labor. The sample size is 563 and 575 households for 2013 and 2014, respectively, which is obtained after excluding the outliers from the analysis. The PCA analysis reduces the input variables into seven factors for 2013 and eight factors for 2014 using the Kaiser (K1) criterion.

After the PCA, Ward linkage (Garson, 2012) is used to group households into clusters while the Duda/Hart index and Calinski-Harabaz criterion are applied as the stopping rule (Costello and Osborne, 2005). We obtain four clusters for both the years.

Cluster	No. of households (%)	Main livelihood activities
2013		
1	280 (49.7)	Farmers
2	146 (25.9)	Natural resource extractors
3	82 (14.5)	Government office with livestock and remittances
4	55 (9.7)	Self-employed and non-agriculture employee
2014		
1	250 (44)	Subsistence farmers and hunting, collecting and logging (HCL) extractors
2	42 (7)	Transition farmers
3	186 (32)	Fish extractors and Government employees
4	97 (17)	Self-employed and non-agriculture employees who receive remittances

Table 3-1	Classification	of livelihoods	by cluster
	••••••••••		

Source: Own calculation.

For 2013, Cluster 1 comprises farmers who constitute about 50% of the households. Cluster 2 constitutes 26% of the households and includes natural resource extractors. These households practice fishing, logging, hunting and collecting. Cluster 3 which forms 14.5% of the households, contains households where at least one of the members is a government official. Also, the cluster receives remittance and transfers from friends and relatives and rears livestock. The last cluster,

cluster 4 includes self-employed and non- agricultural employees. This group has both low skilled and high skilled off farm employees.

For 2014, Cluster 1 constitutes 44% of the households. It includes small farmers who are involved in low-skilled agricultural activities and rear livestock. They also practice hunting, logging and collecting. The second cluster, cluster 2 comprises transition farmers who display characteristics of agricultural commercialization. It forms 7% of the households. Cluster 3 contains households that extract fish and have at least one member who is employed by the Government. Lastly, cluster 4 consists of self-employed and non-agricultural employees and covers 17% of the total households. These households also receive remittances or transfers from friends or family. It can be seen that most of these households (cluster 1 and 3) practice natural resource extraction as incomes from their main occupation are not enough for sustenance.

3.2 **Changes in livelihood strategies**

As evident from the table above, we have a case of identical units and different clusters. This is a unique opportunity as the existing literature does not contain many panel data cluster analysis. Most studies focus on the 'different units, identical cluster' analysis (Van den Berg, 2009). Therefore, our study offers a great chance to identify households that changed clusters and hence livelihood strategies. The information can be used to understand the elements that prompt these shifts. However, it can be seen that though all the clusters are not completely alike, clusters 1 in both the years consist of farmers and also clusters 4 include self-employed and non-agriculture employees in both 2013 and 2014. This makes comparison between the two years more comprehensible.

1	2	3	4	Total
151	26	62	34	273
(55)	(10)	(23)	(12)	(100)
71	3	48	17	139
(51)	(2)	(35)	(12)	(100)
4	7	56	9	76
(5)	(9)	(74)	(12)	(100)
9	4	9	31	53
(17)	(8)	(17)	(58)	(100)
235	40	175	91	541
(43)	(8)	(32)	(17)	(100)
	1 151 (55) 71 (51) 4 (5) 9 (17) 235 (43)	1 2 151 26 (55) (10) 71 3 (51) (2) 4 7 (5) (9) 9 4 (17) (8) 235 40 (43) (8)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Table 3-2 Changes in clusters and their livelihood strategies (No. of households and in %)

Source: Own calculation.

It can be seen in table 3-2 that 55% of the farmers from Cluster 1 in 2013 stayed in the same cluster in 2014. However, it should be mentioned that in 2014, Cluster 1 fairs much worse in terms of all parameters as compared to Cluster 1 in 2013. This implies that the households that did not change clusters contain small scale farmers with the minimum farm areas, least education, investments and transfers. On the other hand, the 10% of the farmers that moved from Cluster 1 in 2013 to the transition farmer cluster (Cluster 2) in 2014 have the highest investment figures out of all the clusters, largest area under cash crop production and better education. They form 65% of Cluster 2 in 2014. Cluster 2 from 2013 witnessed an interesting redistribution in 2014 which has no distinct cluster comprising just natural resource extractors. Rather, these households moved to Cluster 1 and Cluster 3 in 2014 manifesting diversification of livelihood strategies. Diversification has been identified as a self-insurance mechanism by studies such as Barrett and Reardon, 2001. The facts that either income from just one livelihood strategy might not be enough or with the gradual natural resource degradation, households cannot rely completely on income from natural resource extraction, could have prompted this phenomenon.

Cluster 3 and 4 did not witness a big change with most of the households holding on to their existing livelihood practices. However, again there is evidence of diversification.

3.3 **Characteristics of the clusters**

3.3.1 Income and consumption

Table 3-3 illustrates the total income per capita and total consumption per capita for each cluster in 2013 and 2014. It can be seen that the average income and consumption level of households are just around 1000 USD PPP. Although, there is an increase in the income and consumption levels in 2014, yet the figures are much lower than the national income level.

Cluster 2013	1	2	3	4	Whole sample
Total consumption per capita	814	1001	1204	1151	952
Total income per capita	864	1082	1297	1504	1046
Cluster 2014	1	2	3	4	Whole sample
Total consumption per capita	824	963	1146	1149	993
Total income per capita	893	991	1307	1441	1126

Table 3-3 Income and consumption by cluster (in USD PPP)

Source: Own calculation.

In Stung Treng, households mainly participate in income generating-activities such as farming, rearing livestock, natural resources extraction, self-employment, and wage employment. Of these, households whose main activities are not related to agriculture are better off than the remaining groups. Particularly, in 2013 households from Cluster 3 (government officials and livestock feeders), Cluster 4 (self-employed and non-agriculture employees) and Cluster 2 (natural resources extractors), have income and consumption levels higher than the average. Amongst these, Cluster 3 and Cluster 4 boast the highest consumption and income levels, respectively. Only Cluster 1 (farmers) has income and consumption levels under the average.

In 2014, households from group 3 (government official and fishers) and group 4 (self-employee and non-agriculture employment) have income and consumption levels higher than the average with both the highest income and consumption belonging to group 4. Meanwhile, households from the

group of subsistence farmers and forest extractors (Cluster 1) have the lowest income and consumption levels.

3.3.2 Poverty

Poverty is a chronic challenge in rural areas in Cambodia. Although the country has experienced unexpectedly strong progress with a decrease in the poverty rate, the rate of poverty alleviation between areas is unevenly distributed. Meanwhile though the living standard in urban areas is considerably enhanced, the progress in rural areas is still quite modest (Engvall et al., 2007).

	2	013	2014		
Cluster	Income	Consumption	Income	Consumption	
	poverty (%)	poverty (%)	poverty (%)	poverty (%)	
		US \$1.25 PPP			
1	24.6	15.4	28	15.2	
2	28.1	11	26.1	4.7	
3	18.3	2.4	17.2	3.7	
4	14.5	3.6	17.5	2	
Average	23.6	11.2	22.2	6.4	
		US \$2 PPP			
1	46.8	48.6	50.4	49.6	
2	50.7	33.6	40.4	40.4	
3	34.1	19.5	29	20.4	
4	25.5	25.5	32.9	25.7	
Average	43.9	38.2	38.1	34	

Table 3-4 Poverty (headcount ratio) based on income/consumption by cluster

Source: Own calculation.

As illustrated by table 3-4, in 2013 the income and consumption poverty ratios were registered at 23.6% and 11% respectively at \$1.25 PPP. In 2014, there was a significant decrease in the statistics that fell to 22% and 6.4%. The change is also visible at the \$2 PPP level. Regarding livelihood strategies, households from clusters whose main livelihood activities are related to agriculture and natural resources have the highest poverty ratio in terms of both income and consumption.

In 2013, clusters 3 and 4 that contain households with government employees and self-employed members had the lowest income and consumption poverty ratios compared to clusters 1 and 2, farmers and natural resource extractors that had higher poverty ratios. However, there is an anomaly that can be observed. Though income poverty is highest amongst natural resource extractors, consumption poverty is not the highest. This could be attributed to the direct consumption of extracted products. Rayamajhi et al. (2012), in their study of livelihood strategies in Nepal also find that natural products may not always contribute to income but act more as an addition to subsistence.

By comparison, in 2014 the cluster having the highest income and consumption poverty ratio was Cluster 1 with subsistence farmers and forest extractors followed by transition farmers (Cluster 2). Similar to 2013, clusters with government employees and self-employed people had the lowest income and consumption poverty ratio. It must be mentioned that though Cluster 1 and Cluster 2

differ only by a few percentage points in terms of income poverty, the difference is more pronounced in case of consumption.

Interestingly, though the trend in poverty ratio is similar at both the levels of poverty line, the ratio increases significantly, more than 10% for all groups when estimated at \$2 PPP. The effect is more noticeable for clusters whose main livelihoods activities are related to agriculture and natural resources. For example, the consumption poverty headcount ratio of Cluster 1 in 2013 increases more than 30% and for Cluster 2 in 2014 increases more than 37%, compared to their estimates at \$ 1.25 PPP. This indicates that households that are involved in activities pertaining agriculture and natural resources not only have higher poverty rate, but also have higher probability of vulnerability.

4 Selected Livelihood Activities and their Determinants

4.1 Agriculture

4.1.1 Farming

Agriculture is the most important sector in Cambodia. According to a fact sheet published by the FAO (2014), it contributes 37% to the GDP and employs about 67% of the total population. Around 3.7 million hectares of land is cultivated, however, it is argued that better production can be achieved by foreign investment in the sector (Saing et al., 2012). The overall yield per hectare is stated as 3 tons for rice (NSDP, 2013) and the public investment in agriculture has been around 13% between 2009 and 2012 (Siphana et al., 2011).

Though the cultivation of cash crops is expanding, yet 75% of the land is still devoted to rice. Production of rice has witnessed improvement not only in terms of productivity but also with respect to quality. According to the World Bank (2014), there has been a noticeable shift towards high quality white rice. This could work in the favor of the country by increasing its international competitiveness which is struggling with problems such as poor quality seeds, lack of technical know-how and high transportation costs. Cambodia processes only a small part of its rice production and most of it is exported to neighboring countries such as Vietnam and Thailand. The FAO (2014) estimated that in 2013, a record level of 1.2 million tons of paddy was exported from the country.

In Stung Treng, on average every household possesses agricultural land of about 2.8 hectares. However, land security is still a major problem which is evident in our sample as well. In 2013, about 55% of the households had no land documents while 38% had obtained documents from local authorities. The situation witnessed no change in 2014. Another concern is the high dependence on rainfall for irrigation purposes. Only 12% of the surveyed households have proper irrigation facilities in 2014 which is exactly the same as in 2013. Furthermore, in accordance with the national trend, rice still maintains its dominance covering around 40% of the cultivated land. Table 4-1 lends more insight into the agricultural changes with respect to rice.

Parameter	2013	2014	-
Land Size (ha)	1.4	1.7	
Productivity per ha (kg)	2385	1928	
Total Production (kg)	2708	2663	
Production lost after harvest per ha (kg)	41	45	
Consumption (kg)	1909	2040	
Give away (kg)	44	26	
Household processing (kg)	12	18	
Animal Feed (kg)	30	20.1	
Payment in kind (kg)	36	27	
Seed (kg)	159	147	
Sale (directly after harvest) (kg)	399	224	
Sale (3 months later) (kg)	40	75	

Table 4-1 Average agricultural characteristics for rice (mean)

Source: Own calculation.

Though the overall land size has not altered much, there is a significant increase in the land size holding of cluster 4, self-employed and non- agricultural employed households, of 1 ha between 2013 and 2014 (see Appendix 8-3 for cluster wise statistics). Also, Cluster 2 of transition farmers in 2014, on average has bigger land size than their subsistence agricultural counterparts. Furthermore, the production declined by 1.6% between the two years. This could be attributed to the following. Firstly, the productivity decreased from 2.3 tons/ha in 2013 to 1.9 tons/ha in 2014. The biggest decline is observed for Cluster 1, the small scale farmers. Secondly, an increase in loss of production of about 1% was recorded between 2013 and 2014. This also led to a decline in the proportion of produce used for other purposes such as household processing, animal feed and seeds. A noticeable fall of 40% and 27% is seen in terms of rice production being used as a giveaway and payment in kind, respectively. 70% of the produce was consumed in 2013 while the figure augmented to 76% in 2014 with the biggest increase for cluster 4. Additionally, while about 15% of the produce was sold directly after harvest in 2013, only 8% is sold in 2014. The figures for sale after three months of harvest also witness only a slight increase in percentage terms.

4.1.2 Livestock

Livestock can be viewed as a source of protein, additional income and an asset. Therefore, it is no surprise that in Cambodia, about 90% of the total livestock is produced by smallholder farmers (Pen et al., 2009). Rather, in Stung Treng 82% of the households rear livestock such as chickens, ducks, beef cattle, buffalos and pigs. Chickens are more popular and the number of households breeding chicken has witnessed an increase of 15% between 2013 and 2014. This can be due to the minimal investment that is associated with chicken rearing (Burgos et al., 2008). Also, buffalos are seen as assets and are widely reared. The number of households with buffalos increased by 12% in 2014. Generally, oxen are used for fieldwork and buffalos act as savings for the future. Recently, beef cattle has also experienced an increase following a growing demand for beef in neighboring countries such as China. According to Pen et al. (2014), this could provide great opportunity for income expansion to the rural population if they shift from cattle keeping to cattle production. Between 2013 and 2014, 19 more households participated in beef cattle rearing.

Livestock	Rearing Households (No.)		Expenditure/HH (mean USD PPP)		
	2013	2014	2013	2014	
Buffalo	261	293	18	6	
Beef Cattle	106	125	12	7	
Pig (fattening)	187	212	80	62	
Pig (piglet production)	47	56	68	67	
Chicken	324	372	40	17	
Duck	56	69	45	23	

Table 4-2 Livestock rearing in Stung Treng (mean)

Source: Own calculation.

Interestingly, the expenditure incurred by the households on buffalos, beef cattle, chicken and duck decreased manifolds between 2013 and 2014. The same is not visible in case of pigs. Cluster wise, in 2013, Government employees who reared livestock (Cluster 3) had the highest livestock value and, also, realized the highest average sales value. The situation is different in 2014, where Cluster 2 that comprises transition/commercial farmers, has the biggest stock and, also procure the highest average sales value. This corroborated the fact that some households are seeking to commercialize

their livelihood activities. Also, there is a noticeable increase in the overall average sale value. To see statistics for clusters refer to appendix 8-4.

4.2 **Fishing, hunting and logging**

Natural resources play a significant role in supporting livelihoods of rural households in Cambodia. They not only provide households a means for diversifying income and optimizing their labor resources but also act as a valuable "safety net" against adverse shocks (McKenney and Tola, 2002).

Similarly, in Stung Treng a large proportion of the population participates in natural resource extraction, related to both forest and water. The number of households involved in natural resource extraction increased from 79% in 2013 to 89% in 2014. Extracting grounds in Cambodia are generally either open access or community/Government owned. Most of the extraction activities are carried out in open access areas which either have no regulations or ineffective enforcement. For example, 88% of fishing and 98% of small animals are hunted in these areas (Appendix 8-5). This could be termed as one of the potential causes of over exploitation of natural resources that is evident in Cambodia.

4.2.1 Extracted products

A variety of products are extracted in Stung Treng which can be divided into five main groups: (i) fish, (ii) wood, (iii) game, (iv) vegetables and fruits, and (v) small animals.

Product	Output value (USD PPP) (mean)		For sales (USD PPP) (mean)		For consumption (USD PPP) (mean)	
	2013	2014	2013	2014	2013	2014
Fish	1945	1286	1199	630	746	656
Wood	609	910	447	811	162	98
Game	944	1121	651	749	293	472
Vegetables and fruits	296	346	198	257	98	89
Small animals	354	165	256	34	98	132

Table 4-3 Extraction of natural resources

Source: Own calculation.

Table 4-3 shows that on average, fish brings the highest benefit to households followed by game and wood. However, the output value from fishing experienced a significant decrease in 2014 with a fall of nearly one third of the output value in 2013. This could be explained by the over exploitation of water resources especially in the Mekong River which has been impaired in recent years leading to a great decline in fish stock being extracted. More than 80% of the surveyed households reported that there are lesser tree and forest covered areas, while more than 90% of the surveyed households declared that there are lesser fish. Remarkably, on average 495 kg of fish per household was extracted in 2013 whereas only 260 kg of fish was extracted in 2014. In contrast, the output value from extracting wood, vegetables, fruits and game increased between 2013 and 2014.

The extracted products can be used either for consumption or for sale. In 2013, the value of sale exceeded the value of consumption in all types of extracted products. This further emphasizes the

importance of natural resources as a source of cash income for rural households. The pattern remained almost the same in 2014, except in the case of fish and small animals. The value of consumption from extracted fish exceeded the sale value in 2014. An explanation for this could be derived from the fact that generally in times of decreased fish output, consumption is given priority. Therefore, when the quantity of extracted fish decreases, households are more likely to decrease the quantity available for sale.

4.2.2 Income contribution

Table 4-4 highlights the contribution of income from natural resources to the annual household income for each cluster in 2013 and 2014. It accounted for a significant 25% of the total household income in 2013 and decreased slightly to 23% in 2014. However, the decrease was not a consequence of the decline in the absolute value. A growth in total income higher than the growth of natural resources income resulted in this phenomenon.

		Contribution of environmental income (%)			
Cluster	Income (USD PPP (mean)	Total	Water resources (Fish)	Forest resources (Wood, game, fruits and small animals)	
			2013		
1	4138	24	16	8	
2	4720	31	18	13	
3	5920	16	11	5	
4	6519	13	11	2	
Total	4781	25	16	9	
			2014		
1	4308	30	10	20	
2	5515	22	7	9	
3	5897	15	10	5	
4	7292	14	6	9	
Total	5442	23	9	13	

Table 4-4 Contribution of natural resources to annual household income

Source: Own calculation.

The income from natural resources can be derived from two sources: forest and water resources. In 2013, the contribution of water resources was much higher (16%) to the household's annual income than forest resources (9%). However, in 2014 the contribution of water resources decreased to 9% while forest resources formed 13% of the total household income.

With reference to clusters, as expected, natural resources played a more significant role for Cluster 2 (natural resource extractors) in 2013 and Cluster 1 (subsistence farmers and forest extractors) and group 2 (transition farmers) in 2014. Interestingly, despite of having main livelihood activities related to fish extraction, the contribution of natural resources to the annual income of households from Cluster 3 (government office and fish extractors) in 2014 is just 1% higher than the contribution of natural resources in Cluster 4 that contains households with self-employed individuals.

4.3 **Business and wage employment**

The employment structure in Cambodia has not witnessed much change over the past years. Most of the employment is concentrated in rural areas that have lesser productivity. Furthermore, though the unemployment rate is around 3% and most of the poorer households have a job, these are concentrated in the informal sector and, hence, possess neither security nor a decent income (Dalis, 2014). According to the Labor Force Survey that was last conducted in 2012, the labor force participation rate was 69% with 34% and 46% of the population engaged in businesses and wage-salaried jobs, respectively.

4.3.1 Self-employment

In 2013, 22% of the total households were engaged in self-employment. The number increased to 26% in 2014. However, this growth is only limited to the males as the number of business women decreased (refer to Appendix 8-6). The sectors included in the study are: (i) agriculture – various forms of agricultural and livestock trading, (ii) production – industrial production and value upgrading of agricultural products, (iii) trade – retail, transportation and small scale trading, (iv) hotel and food – hotel, restaurants, food processing and sales, and (v) crafts and services – handicrafts, repair shops and service related activities. Table 4-5 shows that while agricultural and production sector businesses experienced a decline between 2013 and 2014, there was an increase in the trade sector (8%) and crafts and services (30%).

Sector	2012 (%)	2014 (%)
Sector	2013 (%)	2014 (%)
Agriculture	8.5	5
Production	8.5	5
Trade	57	62
Food/Hotel	14	13
Crafts & Services	10	13
Others	2	2

Table 4-5 Self-employment by sector

Source: Own calculation.

Breaking the figures into clusters, it can be seen that clusters 1 in 2013 and 2014 that comprise farmers have the same proportion of households (17%) engaged in self-employment. A similar trend is visible for clusters 4 for both the years that contain self-employed and non-agricultural employees where the proportion of households changes only by a percentage point (31% to 32%). In 2014, Cluster 3 has the highest number of self-employed households (43%) while only 7% of the transition farmers group in Cluster 2 are engaged in self-employment.

Sector wise, clusters 3 that include government employees in 2013 and government employees along with fish extractors in 2014, have the highest number of households involved in agriculture related self-employment for 2013 and 2014. However, there has been a significant decline in the numbers. The production sector has also faced the same situation. Clusters 4 contain the highest production related self-employed households for both the years but have decreased. The trade sector along with the other sectors, has witnessed more equitable distribution across the clusters in 2014. However, it can be noticed that though the 'poorer' clusters of 1 and 2 are expanding their business areas, they are still more concentrated in low-skill activities such as rice mill operators, small food store operators or taxi drivers. Sectors requiring medium skill levels such as trade, communication and high skilled activities such as teaching, doctors are still dominated by the richer clusters of 3 and 4. This picture is very similar to that observed in 2013.

When analyzing the socio-demographic figures, the role of education as one of the determinants of self-employment can be easily observed. Both, males and females that are self-employed, have higher levels of education than their non-self-employed counterparts. This is noticeable in 2013 as well as 2014.

4.3.2 Off-farm employment

The number of households engaged in agricultural off farm employment has gone down by 4% between 2013 and 2014. This could be the result of a decrease in mean salary being offered which is much lower compared to other sectors.

Sector	Households (N)		Mean Monthl	y Salary (PPP USD)
	2013	2014	2013	2014
Agriculture	331	371	175	137
Factory	28	19	257	304
Construction	49	89	285	296
Service	40	55	236	245
Public	107	131	208	288
Others	6	3	216	375

Table 4-6 Off-farm em	ployment by sector
-----------------------	--------------------

Source: Own calculation.

As illustrated by table 4-6, the mean salaries for a worker in a factory, or in the construction and service sector, experienced a positive change. At the country level as well, there has been a gradual output growth in the industrial and service sectors while agriculture has remained stagnant (Dalis, 2014). It should be mentioned that 60% of the agricultural off-farm laborers came from the poorest cluster of farmers while households from the richest two clusters, Cluster 3 and Cluster 4 dominated the service and public sectors off-farm employment.

4.4 Migration

Migration is considered as one of the livelihood strategies that households in vulnerable communities pursue to increase and smooth household income, overcome poverty as well as the constraints in their places of origin. Factors such as personal characteristics, wealth, employment opportunities, culture, and political, social, or economic conditions affect the decision to migrate (Nguyen et al., 2014).

In Stung Treng, the number of migrants account for 12% of the whole sample with 60% from the young age group of 15 to 35-year-olds. The average education status of migrants is quite low, with most individuals having no education or primary qualification (Appendix 8-6).

4.4.1 Destination of migrants from Stung Treng

		2013 (%)			2014 (%)		
		Male	Female	Total	Male	Female	Total
Rural	In province	68	52	63	70	61	67
	In another province	14	6	11	8	8	12
Urban	In province	8	15	10	11	12	8
	In another province	3	3	3	3	2	3
	Phnom Penh	2	11	5	2	6	3
	Abroad	5	14	8	6	11	7

Table 4-7 Migrant distribution by destination

Source: Own calculation.

As visible in table 4-7, most of the migrants in Stung Treng moved to rural areas with no significant difference between the two years. This could be explained by the fact that their education status is quite low. Therefore, most of the individuals do not have a great likelihood to obtain work in urban areas where most jobs may require higher education. With reference to gender, it can be seen that the proportion of female migrants in urban areas, Phnom Penh and abroad was higher than the males, which is worth mentioning.

4.4.2 Types of jobs in destination areas

Table 4-8 Types of occupation in destination areas

Types of occupation	2013 (%)	2014 (%)
Agriculture	45	48
Factory	9	5
Construction	9	16
Service	22	16
Public Sector	14	15

Source: Own calculation.

As can be seen in table 4-8, the common trend of job distribution between 2013 and 2014 is that most migrants work in the agriculture sector whereas factories employ the least number of migrants. The low education levels of migrants could also explain this. However, by comparison there has been a remarkable increase in the proportion of the migrants who work in service and construction sectors. Their number increased in construction sector significantly from 9% (one of the least popular occupation) in 2013 to 16%, becoming the second popular occupation in 2014. There was also a considerable increase in the number of migrants employed in the service sector.

4.4.3 Migrant remittances and household welfare

	201	13	2014		
	Non-migrant households	Migrant households	Non-migrant households	Migrant households	
Migration remittances and transfer (USD PPP)	-	330	-	133	
Total household income (USD PPP)	4274	5485	5306	5868	
Total household consumption (USD PPP)	4440	4643	4750	4971	

Table 4-9 Migrant remittances and household welfare

Note: The t-test shows insignificant differences between migrant households and non-migrant households for all welfare indicators.

Source: Own calculation.

In contrast to findings of Jampaklay and Kittisuksathit (2009) in Cambodia, Laos and Myanmar, our study shows that contribution of remittances and transfers to the total income and consumption plays an insignificant role. It equaled just 6% and 7% of the total household income and consumption in 2013 and dropped further to 2% and 3% of the total household income and consumption, respectively in 2014. This is corroborated by the results of t-tests between the welfare indicators of migrant and non-migrant households.

5 Demographic Dividend and Shocks

The livelihood strategies of households are largely influenced by the surroundings, the economic system, socio-political changes, or diverse shocks. As rural households have to adjust to these changes over time, it is pertinent to consider these dynamics over time. In the following, we highlight some of these drivers which influence the livelihood system of rural households. While some of these drivers (such as shocks) are specific to our sample households in Stung Treng, others (e.g. demographic aspects) are more of general natural and apply to Cambodia in general.

5.1 **Demographic dividend**

Since the 1990s, Cambodia has been experiencing a fall in population growth rate and fertility. This coupled with increasing longevity, has led to surplus labor. The annual population growth rate halved to 1.6% while growth rate for labor force has been 2.7%, between 2001 and 2013. This demographic dividend window that opened in 1995 is expected to close around 2045 (Beyene, 2015).

However, this unique opportunity comes with a fair share of problems, most of which could be attributed to its turbulent history. First the schools were closed during the Civil War and then, during the Khmer Rouge, the education system completely collapsed. According to Ayres (1999), classrooms were seen as a symbol of imperialism and, hence, about 75% of the teachers and 96% of higher education students were murdered under the regime. The high and selective mortality not only altered the current socio-demography but also the educational standing of the population (de Walque, 2006).

Since then, the education system has made progress but the youth has not benefitted as expected. The average age of leaving school is 16 years and half of the young population has just primary education. Though unemployment is very low, yet higher unemployment rates are concentrated amongst the educated strata. This implies a clear skill mismatch. Furthermore, the quality of employment is not high for those who are employed.

In Stung Treng, too, the picture is very similar. The mean age of the sample was 25.5 in 2013 and 26 in 2014. More than 50% of the population is between the ages of 16 and 60 for both the years. While there were 891 females and 909 males in the working age bracket in 2013, the number increased to 928 females and 979 males in 2014. Around 65% of these are married and 30% unmarried, for 2013 as well as 2014. There has been a noticeable increase in social involvement in 2014, with 13% more people involved in organizations compared to 2013.

In resonance with the national figures, about 65% of the people between 16-60 years are employed in agriculture. However, between 2013 and 2014, there has been a slight increase in people engaged in non-farm owned business and non-farm employment.

Education received	2013 (%)	2014 (%)
Less than primary	60.5	55.4
Primary/pagoda	23.6	26.7
Secondary school	4.3	5.4
Completed University/Vocational Training	1.50	2

Table 5-1 Educational statistics (16-60 years)

Source: Own calculation.

On the education front, average years of school attended are 5.6 and 5.8 for 2013 and 2014, respectively. 34% of the population between 16 to 60 years had never been to school in 2013 while

the figure decreased to 32% in 2014. Around 23% people had completed primary education in 2013, this increased to 26% in 2014. While there is a visible increase in educational attainment at the lower levels, there is no progress in terms of higher education. Only 10 people in the sample have a Bachelor's degree and 1 person has a Masters. Financial problems are cited as the main cause of leaving school by 43% of the population followed by family problems which forced around 18% people to quit school (2013 and 2014).

According to a recent report published by the ADB (2015), there is ample opportunity but the youth of Cambodia are not well prepared. The Government launched the nation's first National Employment Policy in October 2015 to address the problems of skill gap and educational inadequacy. However, in order to harness the full potential of the demographic dividend, much more is needed. Promoting vocational training and diversifying industrial production could be a start. This would also prove beneficial for natural reserve of the country, which has been exploited due to non-availability of other viable employment options.

5.2 Shocks

Households in Stung Treng have been affected by extreme weather conditions, macroeconomic instability as well as unexpected social adverse events. Due to lack of sustainable food production systems as well as social and economic constraints, they are highly likely to be vulnerable and face difficulties to adapt to such adverse events (Turunen et al., 2011). Adverse events could have been categorized into four main types of shocks: agricultural, economic, health and social shocks. Agricultural shocks refer to floods, droughts, livestock disease and crop pest. Economic shocks refer to rise (fall) in price of input (output), job loss, etc. Health shocks refer to death and illness of household members. Social shocks refer to theft, conflicts with neighbors, ceremony, etc. (Gloede et al., 2015; Van, 2015).

Type of shocks	No. of	shocks	No. of a house	affected holds	Sev	erity	Average los expenditure	s and extra e (PPP USD)
Year	2013	2014	2013	2014	2013	2014	2013	2014
Health	276	383	241	309	1.92	2.36	425	267
Social	69	55	67	51	1.81	2.13	624	760
Agricultural	408	568	301	369	1.93	2.40	389	351
Economic	28	21	23	19	1.57	2.24	393	248
Total	802	1045	421	474	1.92	2.37	423	342

Source: Own calculation.

As can be seen in table 5-2, the number of shocks and the number of households affected increased significantly, except in the case of social and economic shocks. However, the severity of shocks decreased for all types. In both years, agricultural and health shocks were the most common and affected most of the households in this area. On the other hand, social shocks rarely occurred and affected just under 10% of the households, but their severity was perceived to be the most extreme and caused the highest loss to the households.

Cluster	No of shocks	Loss (PPP USD)	Time to recovery (month)
2013			
1	1.5	601	2.5
2	1.2	521	2.3
3	1.2	551	1.5
4	1.1	693	1.6
Whole sample	1.4	582	2.2
2014			
1	1.7	506	2.4
2	1.8	1010	2.5
3	1.7	566	1.9
4	1.6	719	2.3
Whole sample	1.7	598	2.2

Table 5-3 Shocks by cluster

Source: Own calculation.

Table 5-3 illustrates the overview of shocks between main livelihood strategies. In 2013, households in Cluster 1 were affected by the highest number of shocks. This is an expected result as the most common shocks in this area are droughts, floods and storms. On the other hand, despite the fact that Cluster 4 was affected by the smallest number of shocks, the cluster lost the most in terms of value. As social shocks and health shocks that cause the highest loss value occur more often to households in Cluster 4 (appendix 8-7), this can be expected.

In 2014, households in Cluster 2 were most affected by shocks. Not only the number but also the severity and the value of loss were the highest. Amongst the remaining clusters, households in Cluster 4 lost more than households in Cluster 1 and Cluster 3. Also, though households in Cluster 1 and Cluster 3 reported similar number of shocks and the value of loss, the time to recover from shocks of households was higher for Cluster 1.

6 Conclusion

Cambodia is characterized by a high proportion of near poor and hence a huge part of the population faces vulnerability. As households' welfare and vulnerability is highly associated with certain livelihood strategies, an analysis of livelihood activities could entail important results. The current paper endeavors to work in the same direction while contributing to the existing literature. Our case study site, Stung Treng is relatively poorer than other Cambodian provinces despite being endowed with ample natural resources. Using data from 2013 and 2014, we perform separate cluster analyses for both years that enables us to not only determine but also track changes in livelihood strategies. This is a step further compared to existing studies dealing with rural livelihoods that generally do not have panel data.

The first objective of the paper was to identify and describe the changes in rural livelihood strategies in Stung Treng. We find four clusters in both 2013 and 2014 that are not identical but have a common pattern. Clusters that have households engaged in activities pertaining to agriculture or/and natural resources are the poorest while clusters that contain self-employed individuals are the richest in both years. However, in 2014, a unique cluster that contains transition farmers is obtained. These farmers have greater land area and invest much more than any other cluster. With regards to change in strategies, we observe greater movements in the poorer households. While in 2013, we found a distinct cluster for natural resource extractors, these households moved to clusters that practiced additional livelihood strategies in 2014. This indicates a larger diversification. The richer two clusters did not witness considerable shifting.

Income and consumption poverty ratios fell between 2013 and 2014 when calculated at both \$1.25 PPP⁵ and \$2 PPP. Interestingly, the ratios increased significantly for all clusters when the poverty line was changed from \$1.25 to \$2. Again, the vulnerability is higher for households with agriculture and natural resources related livelihoods. This further emphasizes the vulnerability perspective that was mentioned earlier.

The second objective of the paper was to analyze selected livelihood strategies and their determinants. Agriculture in Stung Treng is still highly dependent on natural rainfall with only 12% of the households possessing proper irrigation facilities. There was a decrease in production between 2013 and 2014. This could be attributed to lower productivity or/and greater loss of production. In terms of livestock, chicken and buffaloes are still the most popular with an increase in number of rearing households in 2014. However, the expenditure on buffalos, ducks, beef cattle and chicken declined manifolds between the two years. With reference to natural resource extraction, there was a remarkable decline in the stock of fish extracted by households. Also, the contribution of water resources to annual income was lesser than in 2013. There was a visible shift in labor force from agriculture and production to the services sector. However, the poorer households are still engaged in self-employment that requires low level of skills. The richer clusters continue to dominate high skilled activities.

Additionally, the paper also delves into some specific aspects that are important to understand the livelihood scenario in Stung Treng. Migration has assumed importance with an increase in the total number of migrants between 2013 and 2014. However, still most migrants are single males and move to rural areas. Also, though the households experienced more shocks in 2014 as compared to 2013, the severity was less. This may indicate the better adaptability of households against shocks. Poorer clusters that deal with agriculture and natural resources were affected by higher number of shocks than the richer clusters. The loss and extra expenditure value were more marked in the case of farmers and self-employed households. Furthermore, only 26% of the population between the age

⁵ At \$1.25 PPP, there was a decrease of 1.6 and 4.6 percentage points in income and consumption poverty ratio between 2013 and 2014.

group of 16 to 60 years has completed primary school in Stung Treng that has a mean age of 26. Most of the youth is employed in agriculture.

Overall, although the paper does not have a huge time lag between the two surveyed years, we are able to make some remarkable observations. Firstly, we found a unique group of transition farmers in 2014 (Cluster 2). They are mainly composed of households that moved from the farmers' cluster (Cluster 1) of 2013. However, they differ from their subsistence counterparts in many dimensions. They have higher income and lower consumption poverty than subsistence farmers. Amongst all the obtained clusters, they have the largest area under cash crop production on average and invest the most. This could be attributed to a more 'investment congenial' environment, prompted by improvement in infrastructure that could entail possible expansion opportunities. Also, on average households in this cluster are more educated than small-scale farmers, which could have encouraged engaging in commercial activities. Secondly, households in 2014 display a greater range of diversification. As diversification mitigates risk, this could also explain the decreased severity of shocks in 2014 despite an increase in the number of overall shocks. Lastly, there was a considerable decline in households with self-employment in agriculture and production sector. More households are engaged in crafts and services and trade. All of the above changes can be deemed as positive as there is a gradual movement away from more vulnerable sectors.

However, the analysis also put forth some more concrete evidence with respect to existing issues. Foremost, there has been a considerable decrease in the availability of extracted products, which indicates the worsening of the natural resource problem. For example, the extracted fish output between 2013 and 2014 has also shown a significant decline. This might indicate a potential vulnerability in the future, especially to those who have higher income dependence on natural resources. The deterioration of fish stocks could be attributed to the changes of environment in the Mekong basin. The construction and management of hydropower plants in the Mekong basin changes the volume of water flows, the water quality and the hydrological conditions downstream. This negatively impacts the ecology of the river, fish habitat and migration. In addition, illegal fishing, using destructive fishing methods such as electrocution, poisons or fishing in spawning areas have put fish stock at danger (Coates et al., 2003). Secondly, as emphasized earlier, households that engage in agriculture and natural resources are still the most affected and the longest exposed to shocks. This shows that the vulnerability scenario has not undergone any change. Lastly, the state of education is dismal in the province. Hence, the proposition of achieving a more prosperous and sustainable future for Stung Treng through the educated youth is not in near sight. . However, the promotion of education is most important as it opens up new opportunities for rural households to further diversify their livelihood activities and thus decrease their vulnerability in the longer run.

To sum up, it can be stated that by understanding the various livelihood strategies undertaken by rural households and their changing surroundings, we can garner useful information for rural development initiatives. The current study makes an effort to provide an overall insight with regards to rural Cambodia. Further research is required to analyze the more complex relationship that exists between specific livelihood activities and their determinants.

7 Bibliography

- ADB (2015). Cambodia addressing the skills gap: Employment diagnostic study. Mandaluyong City, ADB.
- Ayres, D. (1999). The Khmer Rouge and education: beyond the discourse of destruction. History of Education, 28(2), 205–218.
- Beyene, K. (2015). Cambodia : Harnessing the Demographic Dividend Lessons from Asian Tigers and other Emerging Asian Economies. DOI: 10.13140/RG.2.1.4771.0567
- Babulo, B., Muys, B., Nega, F., Tollens, E., Nyssen, J., Deckers, J., & Mathijs, E. (2008). Household livelihood strategies and forest dependence in the highlands of Tigray, Northern Ethiopia. Agricultural Systems, 98(2), 147–155.
- Boonsrirat, P. (2014). Common Pool Resources and Rural Livelihoods in Stung Treng Province of Cambodia. Doctoral Dissertations, University of Massachusetts Amherst.
- Burgos, S., Hinrichs, J., Otte, J., Pfeiffer, D., Schwabenbauer, K., & Thieme, O. (2008). Poultry, HPAI and Livelihoods in Cambodia. Mekong Team Working Paper 3, DFID Collaborated Pro-Poor HPAI Risk Reduction Project.
- Bush, S. R. (2005). Fish decline in the Sekong / Se San / Sre Pok River Basin An introduction to its causes and remedies. Victoria, Australia 3065: Oxfam Australia
- Bühler, D., Grote, U., Hartje, R., Ker, B., Do, T.L., Nguyen, L.D., Nguyen, T.T., & Tong, K., 2015. Rural livelihood strategies in Cambodia: evidence from a household survey in Stung Treng. ZEFWorking Paper 137. Center for Development Research, University of Bonn.
- Chhinh, N., & Poch, B. (2012). Climate Change Impacts on Agriculture and Vulnerability as Expected Poverty of Kampong Speu Province, Cambodia. IJERD - International Journal of Environmental and Rural Development, 3(2), 28-37.
- Coates, D., Poeu, O., Suntornratana, U., Thanh Tung, N., & Viravong, S. (2003). Biodiversity and fisheries in the Lower Mekong Basin. Mekong Development Series No. 2, (2), 1–30.
- Costello, A. B., & Osborne, J. W. (2005). Exploratory Factor Analysis. Practical Assessment Research & Evaluation, Vol 10, No. 7.
- Dalis, P. (2014). Links between Employment and Poverty in Cambodia. Working Paper Series No. 92, CDRI Cambodia's leading independent development policy research institute.
- de Sherbinin, A., VanWey, L. K., McSweeney, K., Aggarwal, R., Barbieri, A., Henry, S., ... Walker, R. (2008). Rural household demographics, livelihoods and the environment. Global Environmental Change, 18(1), 38–53.
- de Walque, D. (2006). The Socio-Demographic Legacy of the Khmer Rouge Period in Cambodia. Population Studies : A Journal of Demography., 60(2), 223–231
- Engvall, A., Sjöberg, Ö., & Sjöholm, F. (2007). Poverty in Rural Cambodia: The Differentiated Impact of Linkages, Inputs, and Access to Land. Asian Economic Papers, 7(1129), 74–95.
- FAO. (2014). Cambodia country fact sheet on food and agriculture policy trends. Retrieved January, 20, 2016 from http://www.fao.org/docrep/field/009/i3761e/i3761e.pdf
- Garson, G.D., (2012). Cluster Analysis. Statistical Associates Publishers, Asheboro, NC.
- Gloede, O., Menkhoff,L.&Waibel,H. (2015). Shocks, Individual Risk Attitude, and Vulnerability to Poverty among Rural Households in Thailand and Vietnam. World Development Vol. 71, pp. 54–78
- Hardeweg, B., Klasen, S., &Waibel, H. (2013). Establishing a database for vulnerability assessment. In S. Klasen & H. Waibel (Eds.), Vulnerability to poverty: Theory, measurement, and determinants (pp. 50–79). Basingstoke: Palgrave Macmillan.
- Hing, V., Lun, P., & Phann, D. (2011). Irregular Migration from Cambodia Characteristics, Challenges and Regulatory Approach. CDRI Working Paper Series No. 58.

- Hungry for Life International . (2014). Cambodia Stungtreng: Community development and long-term plan. Retrieved January, 12, 2016 from http://www.hungryforlife.org.
- Jampaklay, A., & Kittisuksathit, S. (2009). Migrant workers' remittances: Cambodia, Lao PDR, and Myanmar. Bangkok, ILO.
- Klasen, S., & Waibel, H., (2015). Vulnerability to poverty in South-East Asia: drivers, measurement, responses, and policy issues. World Development, 71, 1–3.
- Lee, C.C. (2006). Female Labour Migration in Cambodia. Perspectives on Gender and Migration No. E.07.II.F.26, 6-21.
- Ministry of Environment of Cambodia and UNDP Cambodia (2011). Building resilience : The future of Rural Livelihoods in the Face of Climate Change. Cambodia human development report 2011.
- Mai, C. (2003). Making Space and Access in Fisheries Resource Management for Local Communities in Stung Treng Province, Cambodia. RCSD's International Conference on the Politics of the Commons July 11-14, 2003, Chiang Mai
- McKenney, B. & Tola, P. (2002). Natural Resources and Rural Livelihoods in Cambodia. Working Paper No. 23, Cambodia Development Resource Institute.
- Nguyen, D.L., Raabe, K., Grote, U. (2014). Rural–Urban Migration, Household Vulnerability, and Welfare in Vietnam. World Development Vol. 71, pp. 79–93, 2015
- Saing, C.H., Hem, S., Ouch, C., Phann, D.&Pon, D. (2012). Foreign Investment in Agriculture. CDRI Working Paper Series No. 60
- National Institute of Statistics of Cambodia (NIS). (2013). Economic Census of Cambodia, Provincal report: 19 Strung Treng Province. Phnom Penh, Ministry of Planning.
- National Institute of Statistics of Cambodia (NIS). (2009). General Population Census of Cambodia 2008. Phnom Penh, Ministry of Planning.
- Nielsen, J., Rayamajhi, S., Uberhuaga, P., Meilby, H., & Smith-Hall, C., (2013). Quantifying rural livelihood strategies in developing countries using an activity choice approach. Agricultural Economics. 44, 57–71.
- Nguyen, T. T., Do, T. L., Buhler, D., Hartje, R., & Grote, U. (2015). Rural livelihoods and environmental resource dependence in Cambodia. Ecological Economics, 120, 282–295.
- Nguyen, T.T., Bauer, S. & Grote, U. (2016). Does Land Tenure Security Promote Manure Use by Farm Households in Vietnam? Sustainability 2016, 8(2), 178.
- Pen, M., Savage, D. B., LORN, S., & STÜR, W. (2014). Beef Market Chain and Opportunities for Farmers in Kampong Cham Province, Cambodia. IJERD – International Journal of Environmental and Rural Development, (2014) 5-1, 32-37
- Phann, D. (2014). Links between Employment and Poverty in Cambodia. CDRI Working Paper Series No. 92
- Rayamajhi, S., Smith-Hall, C., & Helles, F. (2012). Empirical evidence of the economic importance of Central Himalayan forests to rural households. Forest Policy and Economics, 20, 25–35.
- Siphana, S., Sotharith, C., Vannarith, C. (2011). Cambodia 's Agriculture : Challenges and Prospects. CICP Working paper No. 37.
- Sophal, C., & Acharya, S. (2002). Facing the Challenge of Rural Livelihoods. A Perspective from Nine Villages in Cambodia. CDRI Working Paper No. 25.
- The Forestry Administration (FA). (2010). Cambodia Forestry Outlook Study. APFSOS II Working Paper No. 32
- Tran,V.Q (2015). Household's coping strategies and recoveries from shocks in Vietnam. The Quarterly Review of Economics and Finance 56 (2015) 15–29
- Try, T. & Chambers. M. (2006). Situation Analysis: Stung Treng Province, Cambodia. Mekong Wetlands Biodiversity Conservation and Sustainable Use Programme, Vientiane, Lao PDR.

- Turunen, J., Snäkin, J.-P., Panula-Ontto, J., Lindfors, H., Kaisti, H., Luukkanen, Magistretti, S., & Mang, C. (2011). Livelihood Resilience and Food Security in Cambodia Results from Household Survey.
 Final Report of the research project Knowledge for Development: Creating Rural Resources Database for Sustainable Livelihoods in Cambodia.
- Van den Berg, M. (2009). Household income strategies and natural disasters: Dynamic livelihoods in rural Nicaragua. Ecological Economics, 69, 592–602.
- World Bank (2014). Where Have All The Poor Gone? Cambodia Poverty Assessment 2013. Report No. ACS4545, Second Edition.
- Yusuf, A. A., & Francisco, H. (2009). Climate Change Vulnerability Mapping for Southeast Asia. Economy and Environmental Program for Southeast Asia Edition. Singapore, EEPSEA.

8 Appendix

Veriekle	Cluster				
variable –	1	2	3	4	
No. of members involved in agriculture	2.8	2.5	1.7	.42	
No. of members in natural resource extraction	.3	.8	.2	.1	
No. of members in own business	.05	.1	.3	.9	
No. of members in agri. employment	.52	.13	.09	.15	
No. of members in non-agri. employment	.18	.17	.19	1.26	
No. of members in government sector	.03	.05	.92	.18	
Investments in Agriculture (PPP USD)	986	166	518	229	
Investments in Business (PPP USD)	17.4	174	570	610	
Investments in Natural resource extraction (PPP USD)	143	1310	904	347	
Days per year for Forest Extraction	35.7	127	22.2	18.1	
Days per year for Fishing	70.4	167	48.8	30.8	
Cost for Forest Extraction (PPP USD)	13.6	109	19.8	6.3	
Cost for Fishing (PPP USD)	46.2	301	72.7	18.4	
Land area for Cash Crop (ha)	.73	.34	.57	.27	
Land area for Staple Crop (ha)	1.25	1.24	1.11	.16	
Cost for Farming (PPP USD)	96.8	37.9	44.4	5.8	
Cost for Business (PPP USD)	38.4	43.6	243	3211	
Costs for Livestock (PPP USD)	33	53.6	160	20	
TLU	1.88	2.32	3.96	.79	
Transfers and remittances (PPP USD)	5	20	109	6.1	
Years of education household head	2.7	2.6	6.6	3.6	

Table 8-1 Variable list and summary statistics for clusters 2013

Variable			Cluster	
	1	2	3	4
No. of member involved in agriculture	2.7	2.9	2.2	1.6
No. of members in natural resource extraction	0.7	0.6	0.5	0.3
No. of members in own business	0.08	0.2	0.3	0.7
No. of members in agri. employment	0.5	0.3	0.13	0.06
No. of members in non-agri. employment	0.03	0.08	0.18	1.08
No. of members in government sector	0.03	0.21	0.45	0.14
Investments in Agriculture (PPP USD)	60.97	3696	109	319.6
Investments in Business (PPP USD)	13	1714	25.7	36.7
Investments in Natural resource extraction (PPP USD)	186	637	42	216
Days per year for Forest Extraction	164	107	67.1	58.2
Days per year for Fishing	85.4	70.7	106	44.1
Cost for Forest Extraction (PPP USD)	235.3	147.8	39.6	113
Cost for Fishing (PPP USD)	22.4	67.5	171.9	40.4
Land area for Cash Crop (ha)	0.4	1.4	0.2	0.07
Land area for Staple Crop (ha)	1.12	1.7	1.17	1.15
Cost for Farming (PPP USD)	5.9	9.1	27	5.9
Cost for Business (PPP USD)	588	921	1909	8663
Costs for Livestock (PPP USD)	18.3	28.8	81.1	5.9
TLU	1.4	2.2	2.9	3.4
Transfers and remittances (PPP USD)	14.1	14.9	96.3	11.5
Years of education household head	1.9	4.5	5.3	3.4

Table 8-2 Variables and summary statistics 2014

	2013				2014			
	1	2	3	4	1	2	3	4
Land size(ha)	1.41	1.46	1.95	1.26	1.41	1.86	2.01	2.26
Total Production (kg)	2560	2863	3162	2192	1796	1729	2218	1819
Productivity per ha (kg)	2617	2158	1924	1882	2260	2509	3042	3353
Production loss after harvest per ha (kg)	37	40	65	40	36	37	39	104
Consumption (kg)	1862	1996	1961	1718	1851	2090	2173	2387
Give away (kg)	38	39	91	15	21	4	28	59
Household Processing (kg)	15	3	25	4	14	9	27	13
Animal Feed (kg)	25	31	58	34	14	11	27	33
Payment in kind for labor, machine rental (kg)	35	48	14	35	8	12	48	56
Seed (kg)	144	161	221	190	127	155	162	18
Sale (directly after harvest)	358	465	480	155	114	137	370	331
Sale (3 months later) (in kg)	39	21	103	0	44	32	120	110

Table 8-3 Average agricultural characteristics by cluster for rice (mean)

Cluster	Stock at the Cluster beginning (USD PPP)		Value of animals sold (USD PPP)
		2013	
1	2623	2399	555
2	3415	3647	604
3	5132	4977	1391
4	1566	1260	642
		2014	
1	1790	1984	508
2	3788	2638	1677
3	3137	3367	801
4	3202	3393	822

Table 8-4 Average value (mean) of livestock rearing for households by cluster

Table 8-5 Property rights enforcement status of the extracting grounds

Product	No of HH		Open Access (%)		Community and Government (%)		Others (%)	
	2013	2014	2013	2014	2013	2014	2013	2014
Fish	369	408	88	79	11	21	1	1
Wood	248	444	92	78	6	21	2	1
Game	18	10	95	100	5	0	0	0
Vegetables and fruits	234	413	97	82	3	18	0	1
Small animals	35	89	97	98	3	2	0	0

Table 8-6 Selected socio-demographic characteristics of migrants (%)

		2013			2014	
	Male	Female	Total	Male	Female	Total
		Education L	.evel			
Less than primary	58	45	53	45	42	45
Primary/pagoda	21	34	26	31	32	31
Secondary school	18	21	19	20	22	20
Completed university	0	0	0	2	2	2
Vocational training	3	0	2	2	2	2

Cluster	No. of Health shocks	No. of Social shocks	No. of Agricultural shocks	No. of Economic shocks	No. of shocks	Loss (USD PPP)	Time to recovery (months)
20	13						
1	0.5	0.12	0.8	0.05	1.5	601	2.53
2	0.39	0.09	0.68	0.03	1.2	521	2.34
3	0.40	0.12	0.62	0.04	1.2	551	1.51
4	0.49	0.2	0.33	0.09	1.1	693	1.55
Whole sample	0.46	0.14	0.7	0.04	1.4	582	2.24
2014							
1	0.67	0.06	1.02	0.02	1.7	506	2.4
2	0.64	0.19	0.93	0.02	1.8	1010	2.48
3	0.56	0.1	0.99	0.04	1.7	566	1.88
4	0.71	0.1	0.77	0.06	1.6	719	2.32
Whole sample	0.65	0.09	0.96	0.03	1.7	598	2.23

Table 8-7 Shocks frequency by cluster

ZEF Working Paper Series, ISSN 1864-6638

Center for Development Research, University of Bonn

Editors: Christian Borgemeister, Joachim von Braun, Manfred Denich, Till Stellmacher and Eva Youkhana

- **1.** Evers, Hans-Dieter and Solvay Gerke (2005). Closing the Digital Divide: Southeast Asia's Path Towards a Knowledge Society.
- **2.** Bhuiyan, Shajahan and Hans-Dieter Evers (2005). Social Capital and Sustainable Development: Theories and Concepts.
- **3.** Schetter, Conrad (2005). Ethnicity and the Political Reconstruction of Afghanistan.
- 4. Kassahun, Samson (2005). Social Capital and Community Efficacy. In Poor Localities of Addis Ababa Ethiopia.
- **5.** Fuest, Veronika (2005). Policies, Practices and Outcomes of Demand-oriented Community Water Supply in Ghana: The National Community Water and Sanitation Programme 1994 2004.
- **6.** Menkhoff, Thomas and Hans-Dieter Evers (2005). Strategic Groups in a Knowledge Society: Knowledge Elites as Drivers of Biotechnology Development in Singapore.
- 7. Mollinga, Peter P. (2005). The Water Resources Policy Process in India: Centralisation, Polarisation and New Demands on Governance.
- 8. Evers, Hans-Dieter (2005). Wissen ist Macht: Experten als Strategische Gruppe.
- **8.a** Evers, Hans-Dieter and Solvay Gerke (2005). Knowledge is Power: Experts as Strategic Group.
- **9.** Fuest, Veronika (2005). Partnerschaft, Patronage oder Paternalismus? Eine empirische Analyse der Praxis universitärer Forschungskooperation mit Entwicklungsländern.
- **10.** Laube, Wolfram (2005). Promise and Perils of Water Reform: Perspectives from Northern Ghana.
- **11.** Mollinga, Peter P. (2004). Sleeping with the Enemy: Dichotomies and Polarisation in Indian Policy Debates on the Environmental and Social Effects of Irrigation.
- **12.** Wall, Caleb (2006). Knowledge for Development: Local and External Knowledge in Development Research.
- **13.** Laube, Wolfram and Eva Youkhana (2006). Cultural, Socio-Economic and Political Con-straints for Virtual Water Trade: Perspectives from the Volta Basin, West Africa.
- 14. Hornidge, Anna-Katharina (2006). Singapore: The Knowledge-Hub in the Straits of Malacca.
- **15.** Evers, Hans-Dieter and Caleb Wall (2006). Knowledge Loss: Managing Local Knowledge in Rural Uzbekistan.
- **16.** Youkhana, Eva; Lautze, J. and B. Barry (2006). Changing Interfaces in Volta Basin Water Management: Customary, National and Transboundary.
- **17.** Evers, Hans-Dieter and Solvay Gerke (2006). The Strategic Importance of the Straits of Malacca for World Trade and Regional Development.
- **18.** Hornidge, Anna-Katharina (2006). Defining Knowledge in Germany and Singapore: Do the Country-Specific Definitions of Knowledge Converge?
- **19.** Mollinga, Peter M. (2007). Water Policy Water Politics: Social Engineering and Strategic Action in Water Sector Reform.
- 20. Evers, Hans-Dieter and Anna-Katharina Hornidge (2007). Knowledge Hubs Along the Straits of Malacca.
- **21.** Sultana, Nayeem (2007). Trans-National Identities, Modes of Networking and Integration in a Multi-Cultural Society. A Study of Migrant Bangladeshis in Peninsular Malaysia.
- **22.** Yalcin, Resul and Peter M. Mollinga (2007). Institutional Transformation in Uzbekistan's Agricultural and Water Resources Administration: The Creation of a New Bureaucracy.
- **23.** Menkhoff, T.; Loh, P. H. M.; Chua, S. B.; Evers, H.-D. and Chay Yue Wah (2007). Riau Vegetables for Singapore Consumers: A Collaborative Knowledge-Transfer Project Across the Straits of Malacca.
- 24. Evers, Hans-Dieter and Solvay Gerke (2007). Social and Cultural Dimensions of Market Expansion.
- **25.** Obeng, G. Y.; Evers, H.-D.; Akuffo, F. O., Braimah, I. and A. Brew-Hammond (2007). Solar PV Rural Electrification and Energy-Poverty Assessment in Ghana: A Principal Component Analysis.

- **26.** Eguavoen, Irit; E. Youkhana (2008). Small Towns Face Big Challenge. The Management of Piped Systems after the Water Sector Reform in Ghana.
- 27. Evers, Hans-Dieter (2008). Knowledge Hubs and Knowledge Clusters: Designing a Knowledge Architecture for Development
- **28.** Ampomah, Ben Y.; Adjei, B. and E. Youkhana (2008). The Transboundary Water Resources Management Regime of the Volta Basin.
- **29.** Saravanan.V.S.; McDonald, Geoffrey T. and Peter P. Mollinga (2008). Critical Review of Integrated Water Resources Management: Moving Beyond Polarised Discourse.
- **30.** Laube, Wolfram; Awo, Martha and Benjamin Schraven (2008). Erratic Rains and Erratic Markets: Environmental change, economic globalisation and the expansion of shallow groundwater irrigation in West Africa.
- **31.** Mollinga, Peter P. (2008). For a Political Sociology of Water Resources Management.
- 32. Hauck, Jennifer; Youkhana, Eva (2008). Histories of water and fisheries management in Northern Ghana.
- **33.** Mollinga, Peter P. (2008). The Rational Organisation of Dissent. Boundary concepts, boundary objects and boundary settings in the interdisciplinary study of natural resources management.
- 34. Evers, Hans-Dieter; Gerke, Solvay (2009). Strategic Group Analysis.
- **35.** Evers, Hans-Dieter; Benedikter, Simon (2009). Strategic Group Formation in the Mekong Delta The Development of a Modern Hydraulic Society.
- **36.** Obeng, George Yaw; Evers, Hans-Dieter (2009). Solar PV Rural Electrification and Energy-Poverty: A Review and Conceptual Framework With Reference to Ghana.
- **37.** Scholtes, Fabian (2009). Analysing and explaining power in a capability perspective.
- 38. Eguavoen, Irit (2009). The Acquisition of Water Storage Facilities in the Abay River Basin, Ethiopia.
- **39.** Hornidge, Anna-Katharina; Mehmood UI Hassan; Mollinga, Peter P. (2009). 'Follow the Innovation' A joint experimentation and learning approach to transdisciplinary innovation research.
- **40.** Scholtes, Fabian (2009). How does moral knowledge matter in development practice, and how can it be researched?
- **41.** Laube, Wolfram (2009). Creative Bureaucracy: Balancing power in irrigation administration in northern Ghana.
- **42.** Laube, Wolfram (2009). Changing the Course of History? Implementing water reforms in Ghana and South Africa.
- **43.** Scholtes, Fabian (2009). Status quo and prospects of smallholders in the Brazilian sugarcane and ethanol sector: Lessons for development and poverty reduction.
- **44.** Evers, Hans-Dieter; Genschick, Sven; Schraven, Benjamin (2009). Constructing Epistemic Landscapes: Methods of GIS-Based Mapping.
- **45.** Saravanan V.S. (2009). Integration of Policies in Framing Water Management Problem: Analysing Policy Processes using a Bayesian Network.
- **46.** Saravanan V.S. (2009). Dancing to the Tune of Democracy: Agents Negotiating Power to Decentralise Water Management.
- **47.** Huu, Pham Cong; Rhlers, Eckart; Saravanan, V. Subramanian (2009). Dyke System Planing: Theory and Practice in Can Tho City, Vietnam.
- **48.** Evers, Hans-Dieter; Bauer, Tatjana (2009). Emerging Epistemic Landscapes: Knowledge Clusters in Ho Chi Minh City and the Mekong Delta.
- **49.** Reis, Nadine; Mollinga, Peter P. (2009). Microcredit for Rural Water Supply and Sanitation in the Mekong Delta. Policy implementation between the needs for clean water and 'beautiful latrines'.
- **50.** Gerke, Solvay; Ehlert, Judith (2009). Local Knowledge as Strategic Resource: Fishery in the Seasonal Floodplains of the Mekong Delta, Vietnam

- **51.** Schraven, Benjamin; Eguavoen, Irit; Manske, Günther (2009). Doctoral degrees for capacity development: Results from a survey among African BiGS-DR alumni.
- **52.** Nguyen, Loan (2010). Legal Framework of the Water Sector in Vietnam.
- **53.** Nguyen, Loan (2010). Problems of Law Enforcement in Vietnam. The Case of Wastewater Management in Can Tho City.
- **54.** Oberkircher, Lisa et al. (2010). Rethinking Water Management in Khorezm, Uzbekistan. Concepts and Recommendations.
- **55.** Waibel, Gabi (2010). State Management in Transition: Understanding Water Resources Management in Vietnam.
- **56.** Saravanan V.S.; Mollinga, Peter P. (2010). Water Pollution and Human Health. Transdisciplinary Research on Risk Governance in a Complex Society.
- **57.** Vormoor, Klaus (2010). Water Engineering, Agricultural Development and Socio-Economic Trends in the Mekong Delta, Vietnam.
- **58.** Hornidge, Anna-Katharina; Kurfürst, Sandra (2010). Envisioning the Future, Conceptualising Public Space. Hanoi and Singapore Negotiating Spaces for Negotiation.
- **59.** Mollinga, Peter P. (2010). Transdisciplinary Method for Water Pollution and Human Health Research.
- **60.** Youkhana, Eva (2010). Gender and the development of handicraft production in rural Yucatán/Mexico.
- 61. Naz, Farhat; Saravanan V. Subramanian (2010). Water Management across Space and Time in India.
- **62.** Evers, Hans-Dieter; Nordin, Ramli, Nienkemoer, Pamela (2010). Knowledge Cluster Formation in Peninsular Malaysia: The Emergence of an Epistemic Landscape.
- **63.** Mehmood UI Hassan; Hornidge, Anna-Katharina (2010). 'Follow the Innovation' The second year of a joint experimentation and learning approach to transdisciplinary research in Uzbekistan.
- **64.** Mollinga, Peter P. (2010). Boundary concepts for interdisciplinary analysis of irrigation water management in South Asia.
- **65.** Noelle-Karimi, Christine (2006). Village Institutions in the Perception of National and International Actors in Afghanistan. (**Amu Darya Project Working Paper No. 1**)
- 66. Kuzmits, Bernd (2006). Cross-bordering Water Management in Central Asia. (Amu Darya Project Working Paper No. 2)
- **67.** Schetter, Conrad; Glassner, Rainer; Karokhail, Masood (2006). Understanding Local Violence. Security Arrangements in Kandahar, Kunduz and Paktia. (**Amu Darya Project Working Paper No. 3**)
- **68.** Shah, Usman (2007). Livelihoods in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (**Amu Darya Project Working Paper No. 4**)
- **69.** ter Steege, Bernie (2007). Infrastructure and Water Distribution in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (**Amu Darya Project Working Paper No. 5**)
- **70.** Mielke, Katja (2007). On The Concept of 'Village' in Northeastern Afghanistan. Explorations from Kunduz Province. (Amu Darya Project Working Paper No. 6)
- **71.** Mielke, Katja; Glassner, Rainer; Schetter, Conrad; Yarash, Nasratullah (2007). Local Governance in Warsaj and Farkhar Districts. (**Amu Darya Project Working Paper No. 7**)
- 72. Meininghaus, Esther (2007). Legal Pluralism in Afghanistan. (Amu Darya Project Working Paper No. 8)
- 73. Yarash, Nasratullah; Smith, Paul; Mielke, Katja (2010). The fuel economy of mountain villages in Ishkamish and Burka (Northeast Afghanistan). Rural subsistence and urban marketing patterns. (Amu Darya Project Working Paper No. 9)
- **74.** Oberkircher, Lisa (2011). 'Stay We Will Serve You Plov!'. Puzzles and pitfalls of water research in rural Uzbekistan.
- **75.** Shtaltovna, Anastasiya; Hornidge, Anna-Katharina; Mollinga, Peter P. (2011). The Reinvention of Agricultural Service Organisations in Uzbekistan a Machine-Tractor Park in the Khorezm Region.

- **76.** Stellmacher, Till; Grote, Ulrike (2011). Forest Coffee Certification in Ethiopia: Economic Boon or Ecological Bane?
- **77.** Gatzweiler, Franz W.; Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Marginality. Addressing the roots causes of extreme poverty.
- **78.** Mielke, Katja; Schetter, Conrad; Wilde, Andreas (2011). Dimensions of Social Order: Empirical Fact, Analytical Framework and Boundary Concept.
- **79.** Yarash, Nasratullah; Mielke, Katja (2011). The Social Order of the Bazaar: Socio-economic embedding of Retail and Trade in Kunduz and Imam Sahib
- **80.** Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Innovative business approaches for the reduction of extreme poverty and marginality?
- 81. Ziai, Aram (2011). Some reflections on the concept of 'development'.
- 82. Saravanan V.S., Mollinga, Peter P. (2011). The Environment and Human Health An Agenda for Research.
- **83.** Eguavoen, Irit; Tesfai, Weyni (2011). Rebuilding livelihoods after dam-induced relocation in Koga, Blue Nile basin, Ethiopia.
- **84.** Eguavoen, I., Sisay Demeku Derib et al. (2011). Digging, damming or diverting? Small-scale irrigation in the Blue Nile basin, Ethiopia.
- **85.** Genschick, Sven (2011). Pangasius at risk Governance in farming and processing, and the role of different capital.
- **86.** Quy-Hanh Nguyen, Hans-Dieter Evers (2011). Farmers as knowledge brokers: Analysing three cases from Vietnam's Mekong Delta.
- **87.** Poos, Wolf Henrik (2011). The local governance of social security in rural Surkhondarya, Uzbekistan. Post-Soviet community, state and social order.
- **88.** Graw, Valerie; Ladenburger, Christine (2012). Mapping Marginality Hotspots. Geographical Targeting for Poverty Reduction.
- **89.** Gerke, Solvay; Evers, Hans-Dieter (2012). Looking East, looking West: Penang as a Knowledge Hub.
- **90.** Turaeva, Rano (2012). Innovation policies in Uzbekistan: Path taken by ZEFa project on innovations in the sphere of agriculture.
- **91.** Gleisberg-Gerber, Katrin (2012). Livelihoods and land management in the loba Province in south-western Burkina Faso.
- **92.** Hiemenz, Ulrich (2012). The Politics of the Fight Against Food Price Volatility Where do we stand and where are we heading?
- **93.** Baumüller, Heike (2012). Facilitating agricultural technology adoption among the poor: The role of service delivery through mobile phones.
- **94.** Akpabio, Emmanuel M.; Saravanan V.S. (2012). Water Supply and Sanitation Practices in Nigeria: Applying Local Ecological Knowledge to Understand Complexity.
- **95.** Evers, Hans-Dieter; Nordin, Ramli (2012). The Symbolic Universe of Cyberjaya, Malaysia.
- **96.** Akpabio, Emmanuel M. (2012). Water Supply and Sanitation Services Sector in Nigeria: The Policy Trend and Practice Constraints.
- **97.** Boboyorov, Hafiz (2012). Masters and Networks of Knowledge Production and Transfer in the Cotton Sector of Southern Tajikistan.
- **98.** Van Assche, Kristof; Hornidge, Anna-Katharina (2012). Knowledge in rural transitions formal and informal underpinnings of land governance in Khorezm.
- **99.** Eguavoen, Irit (2012). Blessing and destruction. Climate change and trajectories of blame in Northern Ghana.
- **100.** Callo-Concha, Daniel; Gaiser, Thomas and Ewert, Frank (2012). Farming and cropping systems in the West African Sudanian Savanna. WASCAL research area: Northern Ghana, Southwest Burkina Faso and Northern Benin.

- **101.** Sow, Papa (2012). Uncertainties and conflicting environmental adaptation strategies in the region of the Pink Lake, Senegal.
- **102.** Tan, Siwei (2012). Reconsidering the Vietnamese development vision of "industrialisation and modernisation by 2020".
- 103. Ziai, Aram (2012). Postcolonial perspectives on 'development'.
- **104.** Kelboro, Girma; Stellmacher, Till (2012). Contesting the National Park theorem? Governance and land use in Nech Sar National Park, Ethiopia.
- **105.** Kotsila, Panagiota (2012). "Health is gold": Institutional structures and the realities of health access in the Mekong Delta, Vietnam.
- **106.** Mandler, Andreas (2013). Knowledge and Governance Arrangements in Agricultural Production: Negotiating Access to Arable Land in Zarafshan Valley, Tajikistan.
- **107.** Tsegai, Daniel; McBain, Florence; Tischbein, Bernhard (2013). Water, sanitation and hygiene: the missing link with agriculture.
- **108.** Pangaribowo, Evita Hanie; Gerber, Nicolas; Torero, Maximo (2013). Food and Nutrition Security Indicators: A Review.
- **109.** von Braun, Joachim; Gerber, Nicolas; Mirzabaev, Alisher; Nkonya Ephraim (2013). The Economics of Land Degradation.
- **110.** Stellmacher, Till (2013). Local forest governance in Ethiopia: Between legal pluralism and livelihood realities.
- **111.** Evers, Hans-Dieter; Purwaningrum, Farah (2013). Japanese Automobile Conglomerates in Indonesia: Knowledge Transfer within an Industrial Cluster in the Jakarta Metropolitan Area.
- **112.** Waibel, Gabi; Benedikter, Simon (2013). The formation water user groups in a nexus of central directives and local administration in the Mekong Delta, Vietnam.
- **113.** Ayaribilla Akudugu, Jonas; Laube, Wolfram (2013). Implementing Local Economic Development in Ghana: Multiple Actors and Rationalities.
- **114.** Malek, Mohammad Abdul; Hossain, Md. Amzad; Saha, Ratnajit; Gatzweiler, Franz W. (2013). Mapping marginality hotspots and agricultural potentials in Bangladesh.
- **115.** Siriwardane, Rapti; Winands, Sarah (2013). Between hope and hype: Traditional knowledge(s) held by marginal communities.
- **116.** Nguyen, Thi Phuong Loan (2013). The Legal Framework of Vietnam's Water Sector: Update 2013.
- **117.** Shtaltovna, Anastasiya (2013). Knowledge gaps and rural development in Tajikistan. Agricultural advisory services as a panacea?
- **118.** Van Assche, Kristof; Hornidge, Anna-Katharina; Shtaltovna, Anastasiya; Boboyorov, Hafiz (2013). Epistemic cultures, knowledge cultures and the transition of agricultural expertise. Rural development in Tajikistan, Uzbekistan and Georgia.
- **119.** Schädler, Manuel; Gatzweiler, Franz W. (2013). Institutional Environments for Enabling Agricultural Technology Innovations: The role of Land Rights in Ethiopia, Ghana, India and Bangladesh.
- **120.** Eguavoen, Irit; Schulz, Karsten; de Wit, Sara; Weisser, Florian; Müller-Mahn, Detlef (2013). Political dimensions of climate change adaptation. Conceptual reflections and African examples.
- **121.** Feuer, Hart Nadav; Hornidge, Anna-Katharina; Schetter, Conrad (2013). Rebuilding Knowledge. Opportunities and risks for higher education in post-conflict regions.
- **122.** Dörendahl, Esther I. (2013). Boundary work and water resources. Towards improved management and research practice?
- 123. Baumüller, Heike (2013). Mobile Technology Trends and their Potential for Agricultural Development
- **124.** Saravanan, V.S. (2013). "Blame it on the community, immunize the state and the international agencies." An assessment of water supply and sanitation programs in India.

- **125.** Ariff, Syamimi; Evers, Hans-Dieter; Ndah, Anthony Banyouko; Purwaningrum, Farah (2014). Governing Knowledge for Development: Knowledge Clusters in Brunei Darussalam and Malaysia.
- **126.** Bao, Chao; Jia, Lili (2014). Residential fresh water demand in China. A panel data analysis.
- **127.** Siriwardane, Rapti (2014). War, Migration and Modernity: The Micro-politics of the Hijab in Northeastern Sri Lanka.
- **128.** Kirui, Oliver Kiptoo; Mirzabaev, Alisher (2014). Economics of Land Degradation in Eastern Africa.
- **129.** Evers, Hans-Dieter (2014). Governing Maritime Space: The South China Sea as a Mediterranean Cultural Area.
- **130.** Saravanan, V. S.; Mavalankar, D.; Kulkarni, S.; Nussbaum, S.; Weigelt, M. (2014). Metabolized-water breeding diseases in urban India: Socio-spatiality of water problems and health burden in Ahmedabad.
- **131.** Zulfiqar, Ali; Mujeri, Mustafa K.; Badrun Nessa, Ahmed (2014). Extreme Poverty and Marginality in Bangladesh: Review of Extreme Poverty Focused Innovative Programmes.
- **132.** Schwachula, Anna; Vila Seoane, Maximiliano; Hornidge, Anna-Katharina (2014). Science, technology and innovation in the context of development. An overview of concepts and corresponding policies recommended by international organizations.
- **133.** Callo-Concha, Daniel (2014). Approaches to managing disturbance and change: Resilience, vulnerability and adaptability.
- **134.** Mc Bain, Florence (2014). Health insurance and health environment: India's subsidized health insurance in a context of limited water and sanitation services.
- **135.** Mirzabaev, Alisher; Guta, Dawit; Goedecke, Jann; Gaur, Varun; Börner, Jan; Virchow, Detlef; Denich, Manfred; von Braun, Joachim (2014). Bioenergy, Food Security and Poverty Reduction: Mitigating tradeoffs and promoting synergies along the Water-Energy-Food Security Nexus.
- **136.** Iskandar, Deden Dinar; Gatzweiler, Franz (2014). An optimization model for technology adoption of marginalized smallholders: Theoretical support for matching technological and institutional innovations.
- **137.** Bühler, Dorothee; Grote, Ulrike; Hartje, Rebecca; Ker, Bopha; Lam, Do Truong; Nguyen, Loc Duc; Nguyen, Trung Thanh; Tong, Kimsun (2015). Rural Livelihood Strategies in Cambodia: Evidence from a household survey in Stung Treng.
- **138.** Amankwah, Kwadwo; Shtaltovna, Anastasiya; Kelboro, Girma; Hornidge, Anna-Katharina (2015). A Critical Review of the Follow-the-Innovation Approach: Stakeholder collaboration and agricultural innovation development.
- **139.** Wiesmann, Doris; Biesalski, Hans Konrad; von Grebmer, Klaus; Bernstein, Jill (2015). Methodological review and revision of the Global Hunger Index.
- **140.** Eguavoen, Irit; Wahren, Julia (2015). Climate change adaptation in Burkina Faso: aid dependency and obstacles to political participation. Adaptation au changement climatique au Burkina Faso: la dépendance à l'aide et les obstacles à la participation politique.
- 141. Youkhana, Eva. Postponed to 2016 (147).
- **142.** Von Braun, Joachim; Kalkuhl, Matthias (2015). International Science and Policy Interaction for Improved Food and Nutrition Security: toward an International Panel on Food and Nutrition (IPFN).
- **143.** Mohr, Anna; Beuchelt, Tina; Schneider, Rafaël; Virchow, Detlef (2015). A rights-based food security principle for biomass sustainability standards and certification systems.
- **144.** Husmann, Christine; von Braun, Joachim; Badiane, Ousmane; Akinbamijo, Yemi; Fatunbi, Oluwole Abiodun; Virchow, Detlef (2015). Tapping Potentials of Innovation for Food Security and Sustainable Agricultural Growth: An Africa-Wide Perspective.
- **145.** Laube, Wolfram (2015). Changing Aspirations, Cultural Models of Success, and Social Mobility in Northern Ghana.
- 146. Narayanan, Sudha; Gerber, Nicolas (2016). Social Safety Nets for Food and Nutritional Security in India.

- **147.** Youkhana, Eva (2016). Migrants' religious spaces and the power of Christian Saints the Latin American Virgin of Cisne in Spain.
- **148.** Grote, Ulrike; Neubacher, Frank (2016). Rural Crime in Developing Countries: Theoretical Framework, Empirical Findings, Research Needs.
- **149.** Sharma, Rasadhika; Nguyen, Thanh Tung; Grote, Ulrike; Nguyen, Trung Thanh. Changing Livelihoods in Rural Cambodia: Evidence from panel household data in Stung Treng.

http://www.zef.de/workingpapers.html

ZEF Development Studies

edited by Solvay Gerke and Hans-Dieter Evers

Center for Development Research (ZEF), University of Bonn

Shahjahan H. Bhuiyan Benefits of Social Capital. Urban Solid Waste Management in Bangladesh Vol. 1, 2005, 288 p., 19.90 EUR, br. ISBN 3-8258-8382-5

Veronika Fuest

Demand-oriented Community Water Supply in Ghana. Policies, Practices and Outcomes Vol. 2, 2006, 160 p., 19.90 EUR, br. ISBN 3-8258-9669-2

Anna-Katharina Hornidge Knowledge Society. Vision and Social Construction of Reality in Germany and Singapore Vol. 3, 2007, 200 p., 19.90 EUR, br. ISBN 978-3-8258-0701-6

Wolfram Laube Changing Natural Resource Regimes in Northern Ghana. Actors, Structures and Institutions Vol. 4, 2007, 392 p., 34.90 EUR, br. ISBN 978-3-8258-0641-5

Lirong Liu Wirtschaftliche Freiheit und Wachstum. Eine international vergleichende Studie Vol. 5, 2007, 200 p., 19.90 EUR, br. ISBN 978-3-8258-0701-6

Phuc Xuan To Forest Property in the Vietnamese Uplands. An Ethnography of Forest Relations in Three Dao Villages Vol. 6, 2007, 296 p., 29.90 EUR, br. ISBN 978-3-8258-0773-3 Caleb R.L. Wall, Peter P. Mollinga (Eds.) Fieldwork in Difficult Environments. Methodology as Boundary Work in Development Research Vol. 7, 2008, 192 p., 19.90 EUR, br. ISBN 978-3-8258-1383-3

Solvay Gerke, Hans-Dieter Evers, Anna-K. Hornidge (Eds.) *The Straits of Malacca. Knowledge and Diversity* Vol. 8, 2008, 240 p., 29.90 EUR, br. ISBN 978-3-8258-1383-3

Caleb Wall

Argorods of Western Uzbekistan. Knowledge Control and Agriculture in Khorezm Vol. 9, 2008, 384 p., 29.90 EUR, br. ISBN 978-3-8258-1426-7

Irit Eguavoen

The Political Ecology of Household Water in Northern Ghana Vol. 10, 2008, 328 p., 34.90 EUR, br. ISBN 978-3-8258-1613-1

Charlotte van der Schaaf Institutional Change and Irrigation Management in Burkina Faso. Flowing Structures and Concrete Struggles Vol. 11, 2009, 344 p., 34.90 EUR, br. ISBN 978-3-8258-1624-7

Nayeem Sultana

The Bangladeshi Diaspora in Peninsular Malaysia. Organizational Structure, Survival Strategies and Networks Vol. 12, 2009, 368 p., 34.90 EUR, br. ISBN 978-3-8258-1629-2

Peter P. Mollinga, Anjali Bhat, Saravanan V.S. (Eds.) When Policy Meets Reality. Political Dynamics and the Practice of Integration in Water Resources Management Reform Vol. 13, 2010, 216 p., 29.90 EUR, br., ISBN 978-3-643-10672-8 Irit Eguavoen, Wolfram Laube (Eds.) Negotiating Local Governance. Natural Resources Management at the Interface of Communities and the State Vol. 14, 2010, 248 p., 29.90 EUR, br., ISBN 978-3-643-10673-5

William Tsuma Gold Mining in Ghana. Actors, Alliances and Power Vol. 15, 2010, 256 p., 29.90 EUR, br., ISBN 978-3-643-10811-1

Thim Ly

Planning the Lower Mekong Basin: Social Intervention in the Se San River Vol. 16, 2010, 240 p., 29.90 EUR, br., ISBN 978-3-643-10834-0

Tatjana Bauer

The Challenge of Knowledge Sharing - Practices of the Vietnamese Science Community in Ho Chi Minh City and the Mekong Delta Vol. 17, 2011, 304 p., 29.90 EUR, br., ISBN 978-3-643-90121-7

Pham Cong Huu Floods and Farmers - Politics, Economics and Environmental Impacts of Dyke Construction in the Mekong Delta / Vietnam Vol. 18, 2012, 200 p., 29.90 EUR, br., ISBN 978-3-643-90167-5

Judith Ehlert

Beautiful Floods - Environmental Knowledge and Agrarian Change in the Mekong Delta, Vietnam Vol. 19, 2012, 256 S., 29,90 EUR, br, ISBN 978-3-643-90195-8

Nadine Reis

Tracing and Making the State - Policy practices and domestic water supply in the Mekong Delta, Vietnam Vol. 20, 2012, 272 S., 29.90 EUR, br., ISBN 978-3-643-90196-5 Martha A. Awo Marketing and Market Queens - A study of tomato farmers in the Upper East region of Ghana Vol. 21, 2012, 192 S., 29.90 EUR, br., ISBN 978-3-643-90234-4

Asghar Tahmasebi

Pastoral Vulnerability to Socio-political and Climate Stresses - The Shahsevan of North Iran Vol. 22, 2013, 192 S., 29.90 EUR, br., ISBN 978-3-643-90357-0

Anastasiya Shtaltovna

Servicing Transformation - Agricultural Service Organisations and Agrarian Change in Post-Soviet Uzbekistan Vol. 23, 2013, 216 S., 29.90 EUR, br., ISBN 978-3-643-90358-7

Hafiz Boboyorov

Collective Identities and Patronage Networks in Southern Tajikistan Vol. 24, 2013, 304 S., 34.90 EUR, br., ISBN 978-3-643-90382-2

Simon Benedikter

The Vietnamese Hydrocracy and the Mekong Delta. Water Resources Development from State Socialism to Bureaucratic Capitalism Vol. 25, 2014, 330 S., 39.90 EUR, br., ISBN 978-3-643-90437-9

Sven Genschick

Aqua-`culture'. Socio-cultural peculiarities, practical senses, and missing sustainability in Pangasius aquaculture in the Mekong Delta, Vietnam.

Vol. 26, 2014, 262 S., 29.90 EUR, br., ISBN 978-3-643-90485-0

Farah Purwaningrum

Knowledge Governance in an Industrial Cluster. The Collaboration between Academia-Industry-Government in Indonesia. Vol. 27, 2014, 296 S., 39.90 EUR, br., ISBN 978-3-643-90508-6 Panagiota Kotsila Socio-political and Cultural Determinants of Diarrheal Disease in the Mekong Delta. From Discourse to Incidence Vol. 28, 2014, 376 S., 39.90 EUR, br., ISBN 978-3-643-90562-8

http://www.lit-verlag.de/reihe/zef



Working Paper Series

Authors:	Rasadhika Sharma, Thanh Tung Nguyen, Ulrike Grote, Trung Thanh Nguyen
Contacts:	sharma@iuw.uni-hannover.de, tung.nguyen@iuw.uni-hannover.de, grote@iuw.uni-hannover.de, thanh.nguyen@iuw.uni-hannover.de
Photo:	Ulrike Grote

Published by: Zentrum für Entwicklungsforschung (ZEF) Center for Development Research Walter-Flex-Straße 3 D – 53113 Bonn Germany Phone: +49-228-73-1861 Fax: +49-228-73-1869 E-Mail: zef@uni-bonn.de www.zef.de